Author's response to reviews

Title: Resection of a medial meniscal cyst using a posterior trans-septal approach: a case report

Authors:

Tsuyoshi Ohishi (t-ohishi@ken.ja-shizuoka.or.jp)
Eiji Torikai (eiji1412@hotmail.co.jp)
Daisuke Suzuki (suzu-dai@rc5.so-net.ne.jp)
Tomohiro Banno (tomo-ban.0311@s8.dion.ne.jp)
Yosuke Honda (yosuke@yi8.so-net.ne.jp)

Version: 3 Date: 7 October 2010

Author's response to reviews: see over
Dear Representative of the Independent Editorial Production Team:

Enclosed, please find the revised manuscript entitled “Arthroscopic treatment of a medial meniscal cyst using a posterior trans-septal approach: a case report” with a point-by-point description of the changes I made.

The authors are Tsuyoshi Ohishi, Eiji Torikai, Daisuke Suzuki, Tomohiro Banno, Yosuke Honda. There are 14 pages of text including figure legends and 3 figures.

We would be grateful if the manuscript could be reviewed again and considered for publication in *Sports Medicine Arthroscopy Rehabilitation Therapy Technology*.

The most convenient return address for correspondence is:
Tsuyoshi Ohishi, M.D.
Department of Orthopaedic Surgery
Enshu Hospital
1-1-1 Chuo, Naka-ku, Hamamatsu, Shizuoka 430-0929, Japan
Tel: +81-53-453-1111
Fax: +81-53-401-0081
E-mail address: t-ohishi@ken.ja-shizuoka.or.jp

Sincerely,
Corresponding Author: Tsuyoshi Ohishi M.D., Ph.D.
Answers to the comments.

Major revisions:
1. Thank you for your comment regarding to the abstract. I put the previous version of the abstract by mistake. I put the revised version of the abstract.

Minor revisions:
1. The author details on page 10 was removed.

2. White spaces around the figures were minimized by enlarging each figures as much as possible.

Thank you.

Tsuyoshi Ohishi, MD.